

Make Your Publications Visible.

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre

Foster-McGregor, Neil; Isaksson, Anders; Kaulich, Florian

Working Paper

Importing, Exporting and the Productivity of Services Firms in Sub-Saharan Africa

wiiw Working Paper, No. 98

Provided in Cooperation with:

The Vienna Institute for International Economic Studies (wiiw) - Wiener Institut für Internationale Wirtschaftsvergleiche (wiiw)

Suggested Citation: Foster-McGregor, Neil; Isaksson, Anders; Kaulich, Florian (2013): Importing, Exporting and the Productivity of Services Firms in Sub-Saharan Africa, wiiw Working Paper, No. 98, The Vienna Institute for International Economic Studies (wiiw), Vienna

This Version is available at: https://hdl.handle.net/10419/203963

${\bf Standard\text{-}Nutzungsbedingungen:}$

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.





Working Papers | 98 | March 2013

Neil Foster-McGregor, Anders Isaksson and Florian Kaulich

Importing, Exporting and the Productivity of Services Firms in Sub-Saharan Africa

wiiw Working Papers published since 2010:

- No. 98 N. Foster, A. Isaksson and F. Kaulich: Importing, Exporting and the Productivity of Services Firms in Sub-Saharan Africa, March 2013
- No. 97 N. Foster, A. Isaksson and F. Kaulich: Outward Foreign Direct Investment, Exporting and Firm-Level Performance in Sub-Saharan Africa. March 2013
- No. 96 N. Foster, A. Isaksson and F. Kaulich: Importing, Exporting and Performance in sub-Saharan African Manufacturing Firms. March 2013
- No. 95 S. M. Leitner and R. Stehrer: R&D and Non-R&D Innovators in the Financial Crisis: the Role of Binding Credit Constraints. February 2013
- No. 94 G. Streicher and R. Stehrer: Whither Panama? Constructing a Consistent and Balanced World SUT System including International Trade and Transport Margins. January 2013
- No. 93 M. Holzner: The Determinants of Income Polarization on the Household and Country Level across the EU. September 2012
- No. 92 M. Kelle: Crossing Industry Borders: German Manufacturers as Services Exporters. July 2012
- No. 91 S. M. Leitner, R. Stehrer and B. Dachs: The Gravity of Cross-border R&D Expenditure. July 2012
- No. 90 N. Foster, J. Pöschl and R. Stehrer: Offshoring and the Elasticity of Labour Demand. July 2012
- No. 89 N. Foster, J. Pöschl and R. Stehrer: Manufacturing Productivity: Effects of Service Sector Innovations and Institutions. July 2012
- No. 88 A. Breitwieser and N. Foster: Intellectual Property Rights, Innovation and Technology Transfer: A Survey. June 2012
- No. 87 N. Foster: On the Volume and Variety of Intra-Bloc Trade in an Expanded European Union. June 2012
- No. 86 N. Foster, R. Stehrer and G. de Vries: Offshoring and the Skill Structure of Labour Demand. June 2012
- No. 85 M. Holzner and F. Peci: Measuring the Effects of Trade Liberalization in Kosovo. June 2012
- No. 84 S. M. Leitner and R. Stehrer: Labour Hoarding during the Crisis: Evidence for selected New Member States from the Financial Crisis Survey. June 2012
- No. 83 E. Bekkers and J. Francois: Bilateral Exchange Rates and Jobs. June 2012
- No. 82 E. Bekkers, J. Francois and M. Manchin: Import Prices, Income, and Inequality. June 2012
- No. 81 R. Stehrer: Trade in Value Added and the Valued Added in Trade. June 2012
- No. 80 R. Stehrer, N. Foster and G. de Vries: International spillovers in a world of technology clubs. June 2012
- No. 79 R. Stöllinger: International spillovers in a world of technology clubs. May 2012
- No. 78 S. Leitner and R. Stehrer: Access to Finance and Composition of Funding during the Crisis: A firm-level analysis for Latin American countries. February 2012
- No. 77 E. Bekkers and R. Stehrer: Reallocation Gains in a Specific Factors Model with Firm Heterogeneity. December 2011
- No. 76 M. Holzner and F. Peci: The Impact of Customs Procedures on Business Performance: Evidence from Kosovo. August 2011
- No. 75 C. Hornok: Need for Speed: Is Faster Trade in the EU Trade-Creating? April 2011
- No. 74 S. Leitner and R. Stehrer: Subgroup and Shapley Value Decompositions of Multidimensional Inequality An Application to Southeast European Countries. March 2011
- No. 73 S. M. Leitner and R. Stehrer: Do Exporters Share Part of their Rents with their Employees? Evidence from Austrian Manufacturing Firms. February 2011
- No. 72 S. M. Leitner, J. Pöschl and R. Stehrer: Change begets change: Employment effects of technological and nontechnological innovations – A comparison across countries. January 2011
- No. 71 M. Holzner: Inequality, Growth and Public Spending in Central, East and Southeast Europe. October 2010
- No. 70 N. Foster, J. Pöschl and R. Stehrer: The Impact of Preferential Trade Agreements on the Margins of International Trade. September 2010
- No. 69 L. Podkaminer: Discrepancies between Purchasing Power Parities and Exchange Rates under the Law of One Price: A Puzzle (partly) Explained? September 2010
- No. 68 K. Hauzenberger and R. Stehrer: An Empirical Characterization of Redistribution Shocks and Output Dynamics. August 2010
- No. 67 R. Stöllinger, R. Stehrer and J. Pöschl: Austrian Exporters: A Firm-Level Analysis. July 2010

Neil Foster-McGregor is a research economist at the Vienna Institute for International Economic Studies (wiiw). Anders Isaksson is Industrial Development Officer at UNIDO. Florian Kaulich is Assistant Professor at the Department of Economics, Vienna University of Economics and Business.

This document represents work in progress and is intended to generate comment and discussion. It is not a fully polished publication. The views expressed herein are those of the author(s) and do not necessarily reflects the views of the United Nations Industrial Development Organization.

Neil Foster-McGregor, Anders Isaksson and Florian Kaulich

Importing, Exporting and the Productivity of Services Firms in Sub-Saharan Africa

Contents

Abs	stract	i
1.	Intro	duction1
2.	Data	and Summary Statistics5
3.	Meth	odology8
4.	Resu	ılts11
	4.1.	Main Results11
	4.2	Productivity Differences between Exporters, Importers and Two-Way Traders16
	4.3	Productivity Differences between Export Starters, Continuers and Exiters17
5.		Conclusions
Ref	erenc	es22
Δnr	endix	24

List of Tables

Table 1	Frequency Table	7
Table 2	Summary Statistics	7
Table 3	Mean Comparison Test Results on Demeaned Data	12
Table 4	Foreign Exposure and Firm-Level Productivity – Kolmogorov-Smirnov Test Results	13
Table 5	OLS Results for Exporters, Importers and Two-Way Traders	14
Table 6	Quantile Regression Results for Exporters, Importers and Two-Way Traders	15
Table 7	Productivity Differences between Two-Way Traders and Exporters and Importers Only	17
Table 8	Productivity Differences between Export Starters, Continuers and Exiters	18
Table 9	OLS Results for Starters, Exiters and Continuers	19
Table 10	Quantile Regression Results for Export Starters, Continuers and Exiters	20
Table A1	Number of Firms in Sample by Country	24
Table A2	Number of Firms in Sample by Industry	25

Abstract

In this paper we examine productivity differences between trading and non-trading firms in the services sector using recently collected data on a sample of 19 sub-Saharan African firms. A variety of parametric and non-parametric tests are implemented in order to examine whether exporters, importers and two-way traders perform better than non-traders, and whether there are differences in performance between different types of trading firms. Our results indicate that services firms that are engaged in international trade perform significantly better than those firms that trade on the domestic market only. Two-way traders and exporters only are found to perform better than importers only, with no significant difference in performance found between two-way traders and exporters only. We further present evidence indicating that there is no significant difference in performance between export starters and export continuers, a result consistent with the self-selection hypothesis for African services firms.

Keywords: productivity, imports, exports, services firms

JEL classification: D24, F10, L10

Neil Foster-McGregor, Anders Isaksson and Florian Kaulich

Importing, exporting and the productivity of services firms in sub-Saharan Africa

1. Introduction

A large empirical literature has developed over the last 15 years or so using firm- and plant-level data to consider whether firms with foreign exposure, and internationally trading firms in particular, perform better than domestically oriented firms. Early empirical literature considered whether exporters performed better than non-exporters. There are two alternative – though not necessarily mutually exclusive – explanations as to why exporters may be more productive than non-exporters; namely self-selection and learning-by-exporting. Self-selection of the more productive firms into export markets may occur because there are additional costs associated with selling goods abroad that may include transport, distribution and marketing costs, the cost of personnel with skill to manage foreign networks, and production costs from modifying domestic products for foreign consumption (Fryges and Wagner, 2007). Learning-by-exporting is potentially important since exporting can be an important channel of information flows with overseas buyers sharing knowledge of the latest design specifications and production techniques that might otherwise be unavailable, as well as providing a competitive environment in which efficiency advantages can be obtained.

Since the seminal study of Bernard and Jensen (1995) there have been a large number of research papers that have considered the relationship between exporting and firm-level performance. Despite differences in methodology and country samples the results tend to be fairly consistent, and point to the conclusion that productivity is higher for exporters (see Wagner (2007) and (2012) for recent surveys and Martins and Yang (2009) for a meta-analysis of existing studies). The majority of existing studies conclude in favour of self-selection and against the learning-by-exporting hypothesis, with only a few studies reaching the opposite conclusion (examples being Kraay, 2002; Bigsten et al., 2004; Aw et al., 2000).

While the focus of the early firm-level literature was on the relationship between exports and productivity there are good reasons to believe that importing could also be a significant source of productivity benefits. Capital and intermediate goods imports for example that

Foreign exposure refers to two (not mutually exclusive) activities of firms, namely their trade status (i.e. whether they export or import) and their ownership (i.e. whether they are foreign-owned). In this paper we concentrate on the former, though there exists a literature that also tests whether foreign-owned firms perform better than domestically-owned ones (see Harris, 2002; Harris and Robinson, 2003; Harris and Li, 2007; Yasar and Morrison-Paul, 2007)

In response to such empirical studies theoretical models such as that of Melitz (2003) were developed that provided a rationale for the observed positive relationship between export status and firm productivity, with firms in these models self-selecting into export markets due to sunk costs of exporting.

embody new technologies would be expected to bring in new knowledge that may ultimately enhance a firm's productivity. Imported intermediates may also affect productivity through the use of imported inputs that are of a better quality than domestic counterparts as well as learning spillovers between foreign and domestic firms.³ There may also exist complementarity effects, whereby combining different intermediates creates gains that are more than the sum of their parts, an outcome that could be due to imperfect substitution across goods as in love-of-variety models Cheaper imports may allow firms to produce existing goods using the same inputs as before, but at a lower cost. Importing could also open up new ways of producing existing goods, and even allow entirely new goods to be made. Once again however any impact of importing on performance could be due to either self-selection or learning-by-importing (or both). We would expect there to be fixed costs associated with importing which would support the view that firms self-select into importing4, with high-productivity firms offshoring their production and low-productivity firms limiting themselves to domestic sourcing.⁵ Andersson et al. (2008) argue however that there are strong arguments in favour of a causal impact of importing on productivity. In particular, by importing a firm can exploit global specialisation and use inputs from the technology frontier. Importing intermediates also allows firms to specialise on activities where it has particular strengths.

In response to these arguments and the increasing availability of firm-level data with information on importing a small empirical literature has considered the relationship between importing and firm-level performance. Empirical studies of importing and firm-level productivity now exist for a number of developed countries (Belgium, Denmark, France, Germany, Ireland, Italy, Portugal, Spain, Sweden, the UK and the USA) as well as a small number of transition and developing countries (Chile, Hungary, India, Indonesia, and Poland). The results of studies of the importer premium tend to indicate that importers perform better than non-importers (for a survey of these results see Wagner (2012)). A number of such studies combine the impact of importing and exporting by allowing the impact of international trade to differ depending upon whether the firms are exporters only, importers only or two way traders (see for example Muuls and Pisu, 2009; Andersson et al., 2008; Castellani

-

Indeed, a large empirical literature at the country and industry level has examined the importance of knowledge spillovers through imports and found them to be economically significant both between developed countries, and also from developed to developing countries (for seminal studies see Coe and Helpman (1995) and Coe et al. (1997)).

Such costs may include search costs as firms seek potential foreign suppliers, as well as costs related to the inspection of goods, negotiation and contract formulation, and to acquisition and customs procedures.

See Antras and Helpman (2004) who develop a model similar to Melitz (2003) in which it is assumed that there are fixed costs to importing, and which results in the self-selection of firms into importing.

Much of this recent literature on importing and performance has concentrated on the firm-level effects of offshoring. In addition to allowing firms to acquire inputs at lower costs and to acquire inputs embodying a higher level of technology, the offshoring of production gives firms the opportunity to allocate their resources to the activities where they are most productive, helping to increase specialisation and benefit from economies of scale. Despite such benefits there are also likely to be costs to the firm from offshoring. Such costs may include those related to differences in language, management culture and legal systems, as well as the search costs involved in finding partners in distant and foreign markets.

et al., 2010; Vogel and Wagner, 2010). The results from such studies indicate that the impact of trade on performance tends to be stronger for two way traders followed by importers and exporters, with all three groups performing better than firms not engaged in international trade. Some of the above studies also test for self-selection versus learning by importing effects (for example Vogel and Wagner, 2010), with the results tending to support the self-selection hypothesis.

The vast majority of the empirical literature searching for a relationship between a firm's trade status and firm-level performance use data on manufacturing firms only. Only very recently have papers begun to appear that consider these relationships for services firms, with all such papers considering a small number of developed countries. Wagner (2012, Table 3) reports information on seven studies considering services firms all written or published in 2010 or 2011 and covering six developed countries (France, Germany, Italy, Netherlands, Sweden, United Kingdom).

The reasons for the lack of interest in services firms in the early literature are likely to relate to a lack of data on services firms as well as the presumption that services were in general non-tradable. Services are usually considered to be immaterial, non-storable and highly customised. Moreover, they usually require direct contact between user and provider. Despite this reported services trade has been rising and now comprises a significant portion of overall trade. According to the WTO, the global value of cross-border services exports in 2007 was \$3.3 trillion, or some 20 per cent of world trade in goods and services (WTO, 2008). The share rises to almost 50 per cent if transactions are measured in terms of direct and indirect value added content. If we add sales of services by foreign affiliates to this the value of services trade increases further.

There are also additional problems in conducting this kind of analysis for services firms relating to the measurement of productivity and international trade, though similar problems often arise in the measurement of manufacturing output and trade. It is often very difficult to measure productivity for many service industries since it is often difficult to define the real output of a service sector. Despite this, empirical analyses have constructed measures of both labour and multifactor productivity for services industries (see for example Triplett and Bosworth, 2004). Existing studies of the relationship between firm performance and international trade status for services firms tend to use a measure of labour productivity (i.e. output per worker). While information on importing by services firms is relatively straightforward to provide – especially since such imports need not be imports of services – there is somewhat more confusion with exporting. One of the main characteristics of trade in services is that only a limited number of services can be traded across borders. In fact, a larger number of services require physical contact between consumers and producers in

For a recent study of manufacturing firms in SSA see Foster-McGregor et al. (2012).

⁸ For the past two decades trade in services has grown faster than merchandise trade.

order to allow the transaction to occur. To account for this, the definition of trade in services has been extended to include four different modes of supply through which international transactions in services can occur. Besides the conventional mode (cross-border trade, Mode 1), the definition also includes movements of consumers to the countries where the services are provided (Mode 2), the commercial presence of services enterprises in the countries where the services are consumed (Mode 3), and finally the temporary movement of workers (Mode 4). This classification has been adopted as a framework for current multilateral negotiations under the General Agreement on Trade in Services (GATS) and regional agreements.

In this paper we begin to fill the gap in the existing literature by considering the relationship between services firm's international trade status and their productivity for a sample of 19 sub-Saharan African (SSA) countries. While it is common to assume that services are dominant in developed economies and manufacturing and agriculture dominate the economies of developing countries this is often not the case. Although services do dominate the economies of developed countries, accounting for almost 75 per cent of GDP in OECD countries (see Francois and Hoekman, 2010), they also form a significant component of developing countries' GDP. Francois and Hoekman (2010) for example note that services accounted for 66 per cent of value-added in Latin America in 2007 and that there has been a marked shift in value added towards the service sectors in sub-Saharan Africa, despite lagging growth rates. Massimiliano et al. (2008) note that services constitute over 50 per cent of GDP in low income countries, and that 47 per cent of GDP growth in sub-Saharan Africa over the period 2000-2005 was accounted for by services, compared with 37% and 16% for industry and agriculture respectively.

The current paper uses data from UNIDO's recently completed African Investor Survey (AIS) (see UNIDO, 2012) and a variety of parametric and non-parametric techniques to estimate the productivity effects of exporting, importing and two-way trading. The paper makes a number of contributions to the existing literature. This is the first paper that we are aware of that reports results on the trade-productivity relationship for services firms in SSA. Indeed, with the exception of Bhattacharya et al. (2010) – who consider differences in productivity between domestically-oriented firms, exporters and firms undertaking outward FDI in two sectors in India - this is the first paper that we are aware of that reports results for services firms in low-income countries. The dataset we use also covers more SSA countries and more firms than any other existing dataset that we are aware of, with some of the countries being surveyed for the first time. The current paper is also one of the few that distinguishes between the productivity of exporters, importers and two-way traders in developing countries. Using UNIDO's AIS data we show that services firms that are engaged in international trade perform significantly better than those firms that sell and purchase in the domestic market only. Two-way traders and exporters only are found to perform better than importers only, with no significant difference in performance found between two-way traders and exporters only. We further present evidence indicating that there is no significant difference in performance between export starters and export continuers, a result consistent with the self-selection hypothesis for African services firms.

The remainder of the paper is set out as follows. Section 2 discusses the data employed in our analysis and reports descriptive statistics; Section 3 discusses the empirical methods used in our analysis; Section 4 reports the main results; and Section 5 summarises and concludes.

2. Data and Summary Statistics

The data are drawn from the most recent UNIDO African Investor Survey (AIS) which was conducted over the period 2010-2011 and which surveys over 6,000 manufacturing and services firms in 19 SSA countries. In order to ensure that the interviewed firms accurately represent the countries' economies, the samples were drawn from sampling frames which contained all available information about business activities in the survey countries. Furthermore, the sample was drawn by stratifying the sampling frames along the dimensions of size (10-49, 50-99 or 100+ employees), ownership (domestic or foreign) and sector (ISIC Rev. 3.1 2-digit level), and selecting companies randomly within each stratum. The data were collected mainly via face-to-face interviews between the respondent and a UNIDO enumerator, along with drop and pick in some occasions. The respondents were usually senior managers of the firm or - in the case of foreign ownership - the local subsidiary. After the interview, the data were checked in the country by supervisors and rechecked at UNIDO headquarters. In this paper we use data on the sub-sample of services firms, which gives a final usable sample covers 2,277 firms in 19 countries. In our analysis we consider all countries together rather than reporting separate results for each country since for a number of countries in our sample there are relatively few firm observations (see Table A1 in the appendix), which would make it difficult to obtain reliable estimates of the productivity differences between internationally trading and non-trading firms.

The UNIDO dataset is unique in that it covers a relatively large number of African countries and a large number of firms. As far as we aware, the survey is the largest single survey for Africa in terms of both country and firm coverage. In addition, the survey is current with the survey having been conducted in 2010 and 2011. One drawback of the AIS for empirical work is that the data have a country and industry dimension only, with no time dimension available. This is considered a drawback since in much of the recent regression analysis on firm-level performance researchers have attempted to deal with issues of endogeneity and firm-level heterogeneity through the use of firm fixed effects and matching economet-

_

The countries (and number of firms by country) are reported in the appendix in Table A1, while the service sectors (and the number of firms by industry) are reported in Table A2. These tables also report information on the number of exporters, importers, two-way traders and foreign-owned firms by country and sector.

rics (see Wagner, 2007 and 2012). The use of panel data also allows one to ask whether there are pre-entry differences between export starters and non-exporters to shed light on the issue of whether high performing firms self-select into export markets or whether firms become more productive through exporting. Despite this, the AIS does ask for some historical information. In particular, the survey asks for information on output, employment and export status in the previous period. We use this information below to examine differences in productivity performance for export starters, exiters and continuers, which allows us to say something about whether productivity differences between exporters and non-exporters are the result of self-selection or learning-by-exporting effects.

In our initial analysis we consider differences between firms with and without foreign exposure across a number of performance indicators. In particular, we compare firms by their; (i) the log of labour productivity (defined as the ratio of output to employment); (ii) the log of Total Factor Productivity (TFP); (iii) the log of sales; (iv) the log of the capital-labour ratio (K/L); (v) log employment; (vi) the log of average wages; and (vii) the log of annual average pre-tax profit margin. In the regression analysis that follows later however we follow much of the existing literature and concentrate on a single measure of performance, namely logged labour productivity. We further define three trade status variables for the following categories: (i) exporters only; (ii) importers only; and (iii) two-way traders (i.e. simultaneous exporters and importers). In the productivity comparison for two-way traders we usually make the comparison between two-way traders and non two-way traders, a group which may include exporters and importers. We do however also report results from a comparison in performance between two-way traders and exporters and importers only in the later analysis.

Before reporting results from the formal statistical tests we initially report a number of descriptive statistics for our sample of services firms. Table 1 reports simple frequency data for the full sample, while Table 2 reports summary statistics for our chosen performance measures for all firms, as well as for the different categories of firms (i.e. domestically oriented, exporters, importers, two-way traders). We can see from Table 1 that a much smaller percentage of firms are exporters only (just 4.9%) than importers only (31.1%), with just 5.6 per cent of firms simultaneously exporting and importing. Exporting firms are found to be the exception therefore, with just over 10 per cent of firms exporting. This is consistent with the majority of the existing literature for both developed and developing countries, which also tends to find that exporting is relatively rare. Data reported in Tables A1 and A2 indicate that in our sample of firm exporting is relatively important in Kenya and Madagascar, with importing being particularly relevant in Mozambique and Lesotho along with Malawi, Zambia, Ghana, Ethiopia and Cape Verde. Two-way traders are relatively important in Kenya and Cameroon. By sector, we find that exporting is a relatively frequent activity in computer and related activities, with importing being popular in other services and both wholesale and retail trade. Two-way traders are found to be more commonly found in computers and related activities, air transport and wholesale trade. Just over a third of firms (36%) are classed as being foreign-owned, with 16% and 45% of the sub-sample of foreign firms being exporters and importers respectively – a higher percentage than for the full sample of firms. In the regression analysis below we will take account of the fact that foreign owned firms are more likely to be involved in international trade by including a foreign ownership dummy variable. The frequency table also indicates that only a very small number of firms stopped exporting or began exporting in the year of the survey (0.8% and 1.7% respectively).

Table 1 Frequency Table

Variable	Frequency	Percentage of Total
Total Number of Firms	2,277	100
Exporter only	111	4.87
Importer only	709	31.14
Two-way	128	5.62
Foreign	817	35.88
Foreign exporter	133	5.84
Domestic exporter	106	4.66
Foreign importer	370	16.25
Domestic importer	467	20.51
Export exiter	17	0.75
Export starter	39	1.71
Export continuer	179	7.86

Notes: This table reports simple frequency statistics of firms by type. When splitting countries into foreign and domestic exporters and importers no distinction is made between firms that both export and import. This explains why the number of foreign and domestic exporters exceeds the number of exporters only for example.

Table 2

Summary Statistics

Variable	All firms	Domestically- Oriented	Exporters	Importers	Two-way Traders
Log Output Per Worker	10.22 (10.30)	9.92 (9.94)	10.99 (10.89)	10.49 (10.62)	11.28 (11.43)
Log TFP	6.73 (6.72)	6.68 (6.62)	7.49 (7.54)	6.64 (6.64)	7.10 (7.34)
Log Sales	13.72 (13.80)	13.36 (13.30)	14.85 (15.02)	13.95 (14.06)	15.26 (15.20)
Log K/L	9.08 (9.19)	9.01 (9.16)	9.58 (9.63)	9.06 (9.12)	9.45 (9.62)
Log Employment	3.49 (3.33)	3.43 (3.22)	3.81 (3.83)	3.46 (3.33)	4.02 (3.91)
Log Average Wages	8.14 (8.20)	8.06 (8.08)	8.64 (8.65)	8.18 (8.23)	8.42 (8.59)
Log Profit Rate	2.65 (2.71)	2.68 (2.71)	2.56 (2.77)	2.66 (2.71)	2.32 (2.30)

The table reports the mean value of the performance indicators for all firms, domestically-oriented firms, exporters only, importers and two-way traders. Also reported in brackets are the median values.

Table 2 reports the mean values of the performance variables, with the median also reported in brackets. These initial summary statistics hint at there being differences between

trading firms and non-trading firms as well as between the different types of trading firms. The table indicates that according to both the mean and median values that trading firms perform better than domestically-oriented firms across nearly all performance measures, the major exception being the log of the pre-tax profit margin. The table also indicates that the mean and median values of output per worker, sales and employment are higher for two-way traders than either exporters or importers, results similar to those found elsewhere. Interestingly, the mean and median values of all performance measures (except the profit rate) are higher for exporters than for importers, which is different to that found in many other studies (mainly looking at manufacturing industries).

3. Methodology

In order to test for differences between internationally trading and non-trading services firms as well as between importing, exporting and two-way trading firms we employ a number of statistical methods. We begin by reporting results from a simple comparison of means test, in which we allow the two distributions to have different variances. Such a test concentrates on only one moment of the distribution however, the mean. As such, we also make use of the concept of first order stochastic dominance, which allows one to both compare and rank the entire distributions of – in our case – firm performance. Establishing stochastic dominance requires that the productivity distributions of the three types of firm differ across all moments of the distribution, which thus provides a stricter test of the model than simply comparing mean productivity levels. In particular, we follow the approaches of Deglado et al. (2002) and Girma et al. (2004) and make use of the non-parametric one-and two-sided Kolmogorov-Smirnov (KS tests).

Letting F and G be two cumulative distribution functions, for example the productivity of exporters and non-exporters, then first order stochastic dominance of F relative to G means that F(z) – G(z) must be less or equal to zero for all values of Z, with strict inequality for some Z. This can be tested using the one-sided and two-sided Kolmogorov-Smirnov (KS) test. The two-sided KS statistic tests the hypothesis that both distributions are identical, and the null and alternative hypotheses can be expressed as:

$$H_0$$
: $F(z) - G(z) = 0$ $\forall z \in \Re$
 H_1 : $F(z) - G(z) \neq 0$ for some $z \in \Re$

While the one-sided test can be formulated as:

$$H_0: F(z) - G(z) \le 0 \quad \forall z \in \Re$$

 $H_1: F(z) - G(z) > 0 \quad \text{for some } z \in \Re$

In order to conclude that F stochastically dominates G requires that one can reject the null hypothesis for the two-sided test, but not for the one-sided test. In our analysis below we report results from the one-sided test for both the hypothesis that F dominates G and that G dominates F.

The KS test statistic for the two- and one-sided tests are:

$$KS_2 = \sqrt{\frac{n.m}{N}} \max_{1 \le i \le N} \{F_n(z_i) - G_m(z_i)\}$$

$$KS_1 = \sqrt{\frac{n.m}{N}} \max_{1 \le i \le N} |F_n(z_i) - G_m(z_i)|$$

respectively, where n and m are the sample sizes from the empirical distributions of F and G respectively, and N = n + m.

Finally, we turn to regression analysis which allows us to condition on other factors affecting performance. In particular, we report results from OLS regressions, where in different specifications we include various fixed effects (i.e. sector and country, sector-country interactions) to control for un-modelled country and sector specific differences in performance. We follow much of the existing literature that considers the productivity premium from trade by estimating an initial equation of the following form:

$$\ln Y_{ijk} = \beta_1 \ln EMP_{ijk} + \beta_2 \left(\ln EMP_{ijk}\right)^2 + \beta_3 AGE_{ijk} + \beta_4 K/L_{ijk} + \beta_5 HK_{ijk} + \beta_6 EXP_{ijk} + \beta_7 IMP_{ijk} + \beta_8 TWOWAY_{ijk} + \theta_i + \varphi_j + \varepsilon_{ijk}$$

$$\tag{1}$$

where Y is our measure of firm performance (i.e. labour productivity) in firm k in country i and sector j, EMP is the (logged) number of employees, AGE is firm age in years, K/L is the (logged) capital-labour ratio, HK is a measure of human capital 10 , EXP, IMP and TWOWAY are dummies for exporters, importers, and two-way traders respectively, and θ_i and φ_j are country- and sector-specific effects respectively. In various specifications these latter effects are replaced by sector-country fixed effects, τ_{ij} . We also include in some specifications a dummy variable taking the value one if the firm is foreign-owned (FOREIGN). This allows us to control for the fact that foreign-owned firms are more likely to be engaged in some form of international trade.

The above regression equation is estimated using standard OLS techniques along with the standard within regression when including sector-country fixed effects. Such models seek to estimate the productivity premia at the conditional mean of the productivity distribution. There are reasons to believe however that the impact of exporting and importing is likely to differ across firms. In particular, the recent theoretical literature on trade and productivity (e.g. Melitz, 2003) suggests that firm heterogeneity is to be expected. To account for this possibility therefore we also estimate the above regression model using quantile regression methods. Quantile regression models seek to model the conditional quantile functions, in which the quantiles of the conditional distribution of the dependent variable are expressed as functions of observed covariates (for more information see Buchinsky (1998) and Koenker and Hallock (2001)). The main advantage of quantile regressions is that po-

Defined as the ratio of technical, administrative and sales workers in total employment.

A firm is defined a foreign-owned if more than 10 per cent of the firm's equity is held by foreigners.

For an introduction to quantile regression models see Buchinsky (1998) and Koenker and Hallock (2001).

tentially different solutions at distinct quantiles may be interpreted as differences in the response of the dependent variable to changes in the regressors at various points along the conditional distribution of the dependent variable. The method thus allows one to estimate different parameters on the trade variables for under-achievers (i.e. those at the lower end of the conditional productivity distribution) and over-achievers (i.e. those at the upper end). In addition to allowing for non-linearities in the relationship between a firm's trading status and its performance, quantile regressions have a number of other advantages over OLS. Median regression methods can be more efficient than mean regression estimators in the presence of heteroscedasticity, while when the error term is non-normal, quantile regression estimators may be more efficient than least squares estimators. Quantile regressions are also robust with regard to outlying observations in the dependent variable. In a sample of heterogeneous firms values of some variables are likely to be far away from others. These outliers could be due to reporting errors or to idiosyncratic events and can have a large influence on the coefficients when estimating the regression model by OLS. By using quantile regression methods which are robust to outlying observations we are able to examine the sensitivity of our results to outliers. The quantile regression objective function is a weighted sum of absolute deviations, which gives a robust measure of location, so that the estimated coefficient vector is not sensitive to outlier observations on the dependent variable.

One problem with the use of quantile regression methods in a panel context arises when including a large number of fixed effects, as is the case when we include sector-dummy fixed effects.¹³ In particular, the inclusion of a large number of fixed effects leads to an incidental parameters problem; with a large number of cross-sectional units and a small number of observations for each cross-sectional unit the estimates of the fixed effects are likely to be poor. The poor quality of the estimates of the fixed effects causes the estimates of the main parameters of interest to be badly behaved. Koenker (2004) discusses approaches to deal with such problems, including a class of penalised quantile regression estimators, while Powell (2010) develops an unconditional quantile regression estimator that allows for the inclusion of fixed effects. Both of these approaches are computationally intensive to implement however. Recently, Canay (2011) has introduced an alternative method of estimating quantile regression models with fixed effects that is easy to implement using standard software. The method is based upon the assumption that the fixedeffects in the model act like pure location shift effects, meaning that the fixed effects are constant across quantiles. Given this assumption, Canay proposes the following two-step estimator:

(i) Estimate the standard fixed effects regression at the conditional mean (i.e. the usual within transformation) and using the estimated parameters from this model construct estimates for the individual fixed effects as $\hat{\alpha}_i = \frac{\sum_{t=1}^T (Y_{it} - X_{it}^{'} \hat{\beta}_{\mu})}{T}$, where $\hat{\alpha}_i$ are the estimates

With 19 countries and 24 sectors we have 456 fixed effects to estimate.

- timated fixed effects, Y_{it} is the dependent variable, X_{it} are the explanatory variables, and $\hat{\beta}_u$ are the estimated parameters from the conditional mean regression.
- (ii) Define $\hat{Y}_{it} \equiv Y_{it} \hat{\alpha}_i$ and estimate the quantile regression(s) using this newly defined variable as the dependent variable.

Canay (2011) shows that this estimator is consistent for large T. Canay (2011) also proposes a bootstrap procedure for estimating the variance-covariance matrix for this estimator. The bootstrap method is implemented by drawing with replacement a sample of size NT and computing the two-step estimator as described above. Repeating this a total of B times the estimated bootstrapped variance-covariance matrix at quantile τ is constructed as:

$$\frac{1}{B}\sum_{j=1}^{B} \left(\hat{\beta}_{j}^{*}(\tau) - \bar{\beta}^{*}(\tau)\right) \left(\hat{\beta}_{j}^{*}(\tau) - \bar{\beta}^{*}(\tau)\right)'$$

where $\hat{\beta}_{j}^{*}(\tau)$ are the estimated parameters from the jth bootstrap and the τ th quantile, and $\bar{\beta}^{*}(\tau) = \frac{1}{p} \sum_{j=1}^{p} \hat{\beta}_{j}^{*}(\tau)$.

We adapt this approach to our dataset, which has a country, sector and firm dimension. In our analysis we account for sector-country fixed effects and so follow step 1 above to construct estimates for the sector-country fixed effects and then use these to define the transformed dependent variable for use in step 2. Analogous to the arguments of Canay (2011) the estimator in this case would be consistent as the number of firms increase.¹⁴

4. Results

4.1. Main Results

We begin our comparison of trading and non-trading firms by conducting simple mean comparison tests for exporters versus non-exporters, importers versus non-importers, and two-way traders versus non two-way traders. To account for differences in these performance measures across sectors and countries we de-mean our performance measures, by constructing a variable equal to the logged value of the performance measure minus the mean of the logged value of performance of all firms in the same country and sector. We also use this demeaning procedure when employing the non-parametric KS test below. Results are reported in Table 3. The results in Table 3 indicate that output per worker, TFP, sales, capital stock and average wages are higher for exporters than for non-exporters, with no significant differences found between the two groups for employment

For brevity we choose not to report results when including country and sector fixed effects separately. Given the relatively small number of fixed effects to be included in this case however, it is possible to include them using standard quantile regression methods. These results are available upon request and are qualitatively consistent with those when including country-sector fixed effects.

and the profit rate. The results for importers versus non-importers are found to be similar with importers performing better than non-importers along all performance criteria other than TFP and the profit rate, for which no significant differences are found. For two-way traders versus non two-way traders we again find no significant differences in TFP between the two types of firms, with no significant difference in wages also found in this case. In this case we also find that the mean profit rate is higher in the non-two-way traders is significantly higher than that for two-way traders.¹⁵

Table 3

Mean Comparison Test Results on Demeaned Data

	Mea	n Value	Alternative Hypothesis (p-values)			
	Traders	Non-Traders	Unequal Means	Difference favour- able to traders	Difference favour- able to non-traders	
Exporters versus Non-Expo	orters					
Log Output Per Worker	0.530	-0.062	0.00***	0.00***	1.00	
Log TFP	0.479	-0.032	0.00***	0.00***	1.00	
Log Sales	0.698	-0.100	0.00***	0.00***	1.00	
Log Capital-Labour Ratio	0.485	-0.047	0.00***	0.00***	1.00	
Log Employment	0.138	-0.036	0.12	0.06*	0.94	
Log Wage	0.234	-0.023	0.02**	0.01**	0.99	
Log Profit Rate	-0.055	0.014	0.58	0.71	0.29	
Importers versus Non-Impo	orters					
Log Output Per Worker	0.128	-0.111	0.00***	0.00***	1.00	
Log TFP	0.055	-0.015	0.43	0.22	0.79	
Log Sales	0.260	-0.216	0.00***	0.00***	1.00	
Log Capital-Labour Ratio	0.156	-0.106	0.00***	0.00***	1.00	
Log Employment	0.133	-0.105	0.00***	0.00***	1.00	
Log Wage	0.132	-0.080	0.00***	0.00***	1.00	
Log Profit Rate	0.017	0.007	0.85	0.43	0.58	
Two-way versus Non Two-v	way Traders					
Log Output Per Worker	0.543	0.031	0.00***	0.00***	1.00	
Log TFP	0.092	-0.003	0.56	0.28	0.72	
Log Sales	0.985	-0.059	0.00***	0.00***	1.00	
Log Capital-Labour Ratio	0.300	-0.018	0.03**	0.02**	0.98	
Log Employment	0.451	-0.027	0.00***	0.00***	1.00	
Log Wage	0.155	-0.009	0.11	0.05*	0.95	
Log Profit Rate	-0.157	0.010	0.05*	0.97	0.03**	

While the results reported in Tables 3 would seem to suggest that services firms that trade internationally perform better than those that do not trade internationally the statistics only

-

We also test for differences in the median of our performance measures across these groups using the Stata package 'cendif'. The results are not reported for reasons of brevity, but are largely similar to those using the test of means. The major difference being that the differences in performance between importers and non-importers using the median test are usually not significant (except for output per worker and sales).

look at one moment of the distribution of the performance measures (i.e. the mean). Table 4 therefore reports a similar set of results using the non-parametric Kolmogorov-Sirnov test. The results in Table 4 are fairly consistent with those from the mean comparison test in Table 3. In particular, we find that exporters are found to stochastically dominate non-exporters along all performance criteria except the profit rate, where no significant differences in the two distributions is found. Results for importers versus non-importers are quite similar, though there no significant differences between the two distributions of TFP either. This is also the case for two-way traders, though as with the mean comparison results we find that in the case of the profit rate the distribution for non-two-way traders dominates that of two-way traders.

Table 4

Foreign Exposure and Firm-Level Productivity – Kolmogorov-Smirnov Test Results

	Equality of I	Distribution		Differences favourable to traders		favourable traders
	Statistic	p-value	Statistic	p-value	Statistic	p-value
Exporters vs Non-Exporters						
Log Output Per Worker	0.2787	0.000***	-0.0290	0.838	0.2787	0.000***
Log TFP	0.3088	0.000***	-0.0287	0.899	0.3088	0.000***
Log Sales	0.2537	0.000***	-0.0098	0.980	0.2537	0.000***
Log Capital-Labour Ratio	0.1508	0.017**	-0.0030	0.98	0.1508	0.008***
Log Employment	0.1272	0.066*	-0.0136	0.962	0.1272	0.033**
Log Wage	0.1703	0.005***	-0.0129	0.966	0.1703	0.002***
Log Profit Rate	0.0768	0.708	-0.0768	0.374	0.0599	0.550
Importare ve Neu Importare						
Importers vs Non-Importers	0.0000	0.000***	0.0047	0.070	0.0000	0.000***
Log Output Per Worker	0.0990	0.000***	-0.0047	0.979	0.0990 0.0773	0.000***
Log TFP	0.0773	0.250 0.000***	-0.0231	0.830		0.125 0.000***
Log Sales	0.1471		-0.0058	0.968	0.1471	
Log Capital-Labour Ratio	00859	0.002***	-0.0079	0.944	0.0859	0.001***
Log Employment	0.1077 0.1065	0.000*** 0.000***	-0.0014	0.998 0.955	0.1077 0.1065	0.000
Log Wage Log Profit Rate	0.1065	0.000	-0.0071 -0.0540	0.955 0.101	0.1065	0.439
Log i font itale	0.0040	0.201	0.0040	0.101	0.0020	0.433
Two-Way vs Non-Two-Way Tr	raders					
Log Output Per Worker	0.1704	0.002***	-0.0054	0.993	0.1704	0.001***
Log TFP	0.0986	0.876	-0.0654	0.736	0.0986	0.497
Log Sales	0.2806	0.000***	-0.0068	0.989	0.2806	0.000***
Log Capital-Labour Ratio	0.1408	0.016**	-0.0099	0.977	0.1408	0.008***
Log Employment	0.2129	0.000***	-0.0009	1.000	0.2129	0.000***
Log Wage	0.1266	0.045**	-0.0220	0.892	0.1266	0.022**
Log Profit Rate	0.1611	0.008***	-0.1611	0.004***	0.0282	0.845

Notes: ***, ** and * indicate significance at the 1, 5, and 10 per cent levels.

We now turn to regression analysis which allows us to control for other firm-specific factors when examining the relationship between a firm's trade status and its performance. For brevity, we also concentrate on a single measure of firm performance, namely the log of

labour productivity, in the regression analysis. Table 5 reports regression results from estimating Equation (1) using OLS. The table reports results when including a number of different fixed effects, with Column (1) including no country or sector fixed effects, Column (2) including country and sector fixed effects separately, and Column (3) including sector-country fixed effects. The final three columns report similar results but also include a dummy variable equal to one if the firm is foreign-owned. We control for foreign ownership since foreign-owned firms have been found to perform better than domestically-owned ones and in our dataset are more likely to be internationally trading firms. The results on employment and employment squared are largely consistent with existing literature and indicate that labour productivity rises with firm size, but at a diminishing rate. Firm age is found to have a positive effect on productivity when significant, while the foreign dummy is large positive and significant. The coefficients on the capital-labour ratio and the measure of human capital are also consistently positive and significant as would be expected.

Table 5						
OLS Resul	Its for Expo	rters, Impo	rters and T	wo-Way Tr	aders	
	(1)	(2)	(3)	(4)	(5)	(6)
ln <i>EMP</i>	0.399***	0.458***	0.385***	0.377***	0.445***	0.366***
	(0.112)	(0.114)	(0.133)	(0.108)	(0.111)	(0.131)
ln <i>EMP</i> ²	-0.0608***	-0.0644***	-0.0576***	-0.0609***	-0.0648***	-0.0575***
	(0.0145)	(0.0144)	(0.0171)	(0.0141)	(0.0140)	(0.0167)
AGE	0.0105***	0.00407	0.00298	0.0116***	0.00484*	0.00348
	(0.00273)	(0.00277)	(0.00316)	(0.00265)	(0.00267)	(0.00304)
K/L	0.329***	0.314***	0.300***	0.326***	0.310***	0.296***
	(0.0214)	(0.0221)	(0.0240)	(0.0209)	(0.0216)	(0.0236)
HK	0.0106***	0.00692***	0.00640***	0.0101***	0.00670***	0.00609***
	(0.00104)	(0.00112)	(0.00127)	(0.00103)	(0.00110)	(0.00125)
EXP	0.759***	0.646***	0.638***	0.676***	0.566***	0.548***
	(0.128)	(0.121)	(0.135)	(0.127)	(0.120)	(0.135)
IMP	0.571***	0.346***	0.342***	0.513***	0.275***	0.260***
	(0.0705)	(0.0780)	(0.0860)	(0.0700)	(0.0768)	(0.0845)
ΓWO	1.231***	0.874***	0.839***	1.076***	0.727***	0.674***
	(0.152)	(0.148)	(0.163)	(0.153)	(0.146)	(0.158)
FOREIGN				0.503***	0.508***	0.529***
				(0.0693)	(0.0691)	(0.0760)
$H_0: EXP = IMP$	2.00	5.49***	4.34**	1.56	5.23**	4.22**
$H_0: EXP = TWO$	6.40**	1.74	1.11	4.69**	0.91	0.46
$H_0: IMP = TWO$	18.08***	13.11***	9.60***	13.11***	9.89***	7.11***
Sector and Country Fixed Effects	No	Yes	No	No	Yes	No
Sector-Country Fixed Effects	No	No	Yes	No	No	Yes
Observations	2,125	2,125	2,125	2,125	2,125	2,125
F-Statistic	82.28***	25.09***	5.05***	81.29***	26.41***	5.36***
R-squared	0.255	0.372	0.458	0.273	0.389	0.473
Robust standard errors in parenthese	es					
*** p<0.01, ** p<0.05, * p<0.1						

Turning to the trade variables we find coefficients on the exporter, importer and two-way dummy variables that are large positive and significant in all cases. There seems to be a pattern in the results with importers having the smallest coefficient and two-way traders the largest. The size of the estimated productivity premia, particularly for two-way traders, are often found to be large, though the size of the coefficients tend to fall when the foreign ownership dummy is included and when either sector and country or sector-country fixed effects are included. The results suggest that the premia for two-way traders ranges from a low of 96 per cent to a high of 242 per cent, with the premia from exporting ranging from 73 to 113 per cent and the premia from importing ranging from 30 to 77 per cent. Tests of the equality of these coefficients indicate that there tends not to be a significant difference in productivity between exporters and two-way traders, but that there are usually significant differences between importers and both exporters and two-way traders, indicating that exporters and two-way traders perform significantly better than importers. This is something we return to briefly below.

Table 6					
Quantile	Regression Res	ults for Exporte	rs, Importers ar	nd Two-Way Tr	aders
	(1) 10 th	(2) 30 th	(3) 50 th	(4) 70 th	(5) 90 th
ln <i>EMP</i>	0.304*	0.506***	0.426***	0.260**	0.117
	(0.165)	(0.113)	(0.141)	(0.110)	(0.200)
ln <i>EMP</i> ²	-0.0502**	-0.0707***	-0.0627***	-0.0452***	-0.0352
	(0.0217)	(0.0143)	(0.0178)	(0.0137)	(0.0250)
AGE	0.0167***	0.0138***	0.0118***	0.0113***	0.0106**
	(0.00448)	(0.00289)	(0.00338)	(0.00251)	(0.00420)
K/L	0.389***	0.340***	0.342***	0.315***	0.265***
	(0.0349)	(0.0210)	(0.0256)	(0.0198)	(0.0389)
HK	0.00641***	0.00896***	0.0105***	0.0112***	0.0111***
	(0.00181)	(0.00121)	(0.00155)	(0.00118)	(0.00196)
EXP	1.027***	0.719***	0.567**	0.595***	0.532*
	(0.246)	(0.178)	(0.222)	(0.168)	(0.276)
IMP	0.534***	0.627***	0.675***	0.528***	0.398***
	(0.120)	(0.0839)	(0.105)	(0.0792)	(0.133)
TWO	0.891***	0.954***	1.189***	1.269***	1.239***
	(0.240)	(0.165)	(0.206)	(0.156)	(0.265)
FOREIGN	0.342***	0.456***	0.417***	0.503***	0.641***
	(0.115)	(0.0796)	(0.100)	(0.0765)	(0.129)
$H_0: EXP = IMP$	3.77*	0.25	0.22	0.15	0.22
$H_0: EXP = TWO$	0.18	1.04	4.69**	9.62***	3.87**
$H_0: IMP = TWO$	2.08	3.73*	5.90**	21.49***	9.54***
Psuedo R ²	0.180	0.184	0.168	0.155	0.124
Observations	2,125	2,125	2,125	2,125	2,125
Standard errors in paren	theses				
*** p<0.01, ** p<0.05, * p	><0.1				

The premia are calculated from the estimated coefficients on the trade dummies as $100(e^{\beta}-1)$, where β is the estimated coefficient.

Table 6 reports results from estimating the regression model using quantile regression and the fixed effects quantile estimator of Canay (2011) in particular. Results are reported for the 10th, 30th, 50th (i.e. median), 70th and 90th percentiles of the conditional productivity distribution. Results on the control variables are largely similar to those when using OLS in terms of sign and significance, though firm size is not found to be a significant determinant of productivity in the biggest over-achievers (i.e. firms at the 90th percentiles). Coefficients on the trade dummies are generally consistent with those from the OLS results, with positive and significant coefficients found in all cases. The ranking of the coefficients by size also tends to be consistent with the OLS results, with two-way traders performing better than exporters only, who in turn perform better than importers only (the exception being at the 10th percentile where exporters are the best performing firms). Once again, the differences in the coefficients are generally significant for the comparison between importers only and two-way traders, with significant differences also found between two-way traders and exporters at higher quantiles. Interestingly, in the case of exporters only and importers only the coefficients on the trade status dummies tend to be largest for under-achievers, declining as we move to higher quantiles. The reverse is the case for two-way traders however, with the coefficients on this variable increasing as we move to higher quantiles.

4.2 Productivity Differences between Exporters, Importers and Two-Way Traders

In the more recent literature on international trade and firm-level performance attention has been paid to the issue of whether there are productivity differences between different types of trading firms, and in particular between exporters only, importers only and two-way traders (see Wagner, 2012). Results from existing studies tend to indicate a clear ranking in terms of performance by trading type, with two-way traders having the highest levels of productivity, followed by importers only and then exporters only. In the regression results reported above we find somewhat different results with importers only having the smallest productivity premium, and no significant differences found between the productivity premium for two-way traders and exporters only. We now further examine this issue by reporting results from the KS test, which allows us to compare the entire distribution.

Table 7 reports the results from the KS test for logged labour productivity as well as the other performance indicators considered above. The table reveals that there are few significant differences in the distributions of the performance measures between exporters and two-way traders. Only in the case of TFP and the profit rate do we find significant differences, which in both cases favour the exporting group. In the case of importers versus two-way traders significant differences are found for sales, employment and the profit rate, the former two of which favour two-way traders and the latter favouring importers. In the case of exporters versus importers we observe that there are significant differences in the distributions of output per worker, TFP, sales and wages, all of which favour exporters. Overall, the results tend to suggest that there are fewer differences between the trading

groups than is often found, particularly with respect to our main variable of interest, labour productivity. Moreover, exporters are found to perform better than importers across many criteria, a result that is not often found in the literature, while they are found to perform as well and sometimes better than two-way traders, a group that is often found to perform better than others.

Table 7

Productivity Differences between Two-Way Traders and Exporters and Importers Only

Group 1 vs. Group 2	Equality of Distribution			Differences favourable to Group 1		Differences favourable to Group 2	
	Statistic	p-value	Statistic	p-value	Statistic	p-value	
Exporters vs Two Way Tra	nders						
Log Output Per Worker	0.1416	0.182	0.1010	0.296	-0.1416	0.091*	
Log TFP	0.2797	0.046**	0.0251	0.970	-0.2797	0.023**	
Log Sales	0.1102	0.463	0.1102	0.235	-0.0402	0.824	
Log Capital-Labour Ratio	0.0538	0.995	0.0513	0.730	-0.0538	0.708	
Log Employment	0.1347	0.229	0.1347	0.115	0.0000	1.000	
Log Wage	0.0674	0.952	0.0090	0.990	-0.0674	0.585	
Log Profit Rate	0.2014	0.036**	0.0530	0.757	-0.2014	0.018**	
	_						
Importers vs Two Way Tra							
Log Output Per Worker	0.1164	0.104	0.1164	0.052*	-0.0078	0.987	
Log TFP	0.0983	0.921	0.0602	0.796	-0.0983	0.544	
Log Sales	0.2063	0.000***	0.2063	0.000***	-0.0078	0.987	
Log Capital-Labour Ratio	0.093	0.301	0.0933	0.151	-0.0194	0.921	
Log Employment	0.1583	0.008***	0.1583	0.004***	-0.0015	0.999	
Log Wage	0.1136	0.128	0.1136	0.064*	-0.0361	0.758	
Log Profit Rate	0.1489	0.029**	0.0297	0.845	-0.1489	0.015**	
Exporters vs Importers							
Log Output Per Worker	0.2180	0.000***	-0.0398	0.738	0.2180	0.000***	
Log TFP	0.2911	0.000***	-0.0373	0.866	0.2911	0.000***	
Log Sales	0.1675	0.009***	-0.0185	0.937	0.1675	0.005***	
Log Capital-Labour Ratio	0.0958	0.344	-0.0104	0.980	0.0958	0.173	
Log Employment	0.0680	0.767	-0.0680	0.412	0.0669	0.424	
Log Wage	0.1491	0.029**	-0.0139	0.964	0.1491	0.014**	
Log Profit Rate	0.0800	0.710	-0.0626	0.548	0.0800	0.374	

Notes: ***, ** and * indicate significance at the 1, 5, and 10 per cent levels.

4.3 Productivity Differences between Export Starters, Continuers and Exiters

While there are only limited data on past performance and status in the AIS, the dataset does include information on export status in the previous period. From this data we are able to construct variables indicating whether firms began exporting, continued exporting or stopped exporting in the survey year. We are thus able to ask whether there are differences in performance between export starters, continuers and exiters, which allows us to

say something on the question of whether productivity differences between exporters and non-exporters is due to self-selection or learning-by-exporting in our sample of countries. If learning-by-exporting is not relevant, with the productivity premium of exporters being due to the self-selection of more productive firms into exporting, then we would expect there to be no significant differences in productivity between export starters and continuers.

Table 8 Productivity Differences between Export Starters, Continuers and Exiters

Group 1 vs. Group 2	Equality of Distribution			favourable oup 1		favourable oup 2
	Statistic	p-value	Statistic	p-value	Statistic	p-value
Export Starter vs Continui	ing Exporter					
Log Output Per Worker	0.0959	0.934	-0.0959	0.560	0.0453	0.879
Log TFP	0.1677	0.848	-0.1179	0.691	0.1677	0.473
Log Sales	0.1241	0.718	-0.1241	0.379	0.1094	0.471
Log Capital-Labour Ratio	0.1188	0.766	-0.0089	0.995	0.1188	0.412
Log Employment	0.1143	0.805	-0.1143	0.439	0.0443	0.884
Log Wage	0.0839	0.980	-0.0839	0.643	0.0766	0.692
Log Profit Rate	0.3062	0.016**	-0.0321	0.948	0.3062	0.008***
Export Exiter vs Continuir	ng Exporter					
Log Output Per Worker	0.2860	0.137	-0.2860	0.069*	0.0055	0.999
Log TFP	0.1620	1.000	-0.1544	0.799	0.1620	0.781
Log Sales	0.4417	0.003***	-0.4417	0.002***	0.0055	0.9999
Log Capital-Labour Ratio	0.2398	0.333	-0.0370	0.958	0.2398	0.167
Log Employment	0.3625	0.027**	-0.3625	0.013**	0.0155	0.992
Log Wage	0.1683	0.771	-0.1441	0.525	0.1683	0.415
Log Profit Rate	0.1905	0.740	-0.0522	0.932	0.1905	0.394
Export Exiter vs Export St	arter					
Log Output Per Worker	0.2865	0.268	-0.2865	0.135	0.0322	0.975
Log TFP	0.2875	0.911	-0.2875	0.533	0.1500	0.842
Log Sales	0.4883	0.006***	-0.4883	0.003***	0.0088	0.998
Log Capital-Labour Ratio	0.2198	0.622	-0.0975	0.800	0.2198	0.321
Log Employment	0.3538	0.094*	-0.3538	0.047**	0.026	0.983
Log Wage	0.2152	0.648	-0.2152	0.337	0.1347	0.653
Log Profit Rate	0.2488	0.589	-0.2488	0.303	0.1636	0.597

Notes: ***, ** and * indicate significance at the 1, 5, and 10 per cent levels.

To examine whether there are differences in productivity between export starters, continuers and exiters we follow the same steps as above. Table 8 reports the results from the KS test. The results indicate that there are in general no significant differences in the distributions of the performance measures between continuing exporters and export starters, the exception being the profit rate for which export starters stochastically dominate. There are also few significant differences found between the distributions of export exiters and the other two groups. Significant differences exist for sales and employment when considering the comparison with export exiters and both starters and continuers, and in all cases favour

starters of continuers. Overall however, we find few differences between the three groups of exporting firms.

Table 9 OLS	Results fo	or Starters, I	Exiters and	Continuers	3	
	(1)	(2)	(3)	(4)	(5)	(6)
ln <i>EMP</i>	0.402***	0.459***	0.387***	0.379***	0.446***	0.367***
	(0.112)	(0.113)	(0.133)	(0.108)	(0.110)	(0.130)
ln <i>EMP</i> ²	-0.0611***	-0.0644***	-0.0578***	-0.0612***	-0.0648***	-0.0577**
	(0.0145)	(0.0144)	(0.0170)	(0.0141)	(0.0140)	(0.0166)
AGE	0.0102***	0.00370	0.00252	0.0114***	0.00450*	0.00306
	(0.00274)	(0.00278)	(0.00317)	(0.00266)	(0.00268)	(0.00305
KL	0.329***	0.315***	0.303***	0.326***	0.311***	0.298***
	(0.0215)	(0.0221)	(0.0240)	(0.0210)	(0.0217)	(0.0235)
HK	0.0106***	0.00688***	0.00629***	0.0101***	0.00665***	0.00598*
	(0.00104)	(0.00112)	(0.00126)	(0.00103)	(0.00110)	(0.00125
START	0.759***	0.427*	0.302	0.703***	0.364	0.238
	(0.252)	(0.240)	(0.252)	(0.247)	(0.241)	(0.248)
CONTINUE	0.747***	0.727***	0.781***	0.657***	0.644***	0.685***
	(0.135)	(0.122)	(0.135)	(0.134)	(0.120)	(0.132)
EXIT	-0.143	-0.320	-0.550	-0.215	-0.383	-0.614
	(0.462)	(0.474)	(0.517)	(0.460)	(0.467)	(0.499)
IMP	0.564***	0.339***	0.339***	0.507***	0.270***	0.258***
	(0.0704)	(0.0781)	(0.0862)	(0.0699)	(0.0769)	(0.0847)
TWO	0.497***	0.193	0.127	0.424**	0.124	0.0525
	(0.184)	(0.167)	(0.184)	(0.183)	(0.164)	(0.178)
FOREIGN	()	(/	(/	0.506***	0.511***	0.530***
				(0.0693)	(0.0690)	(0.0760)
$H_0: START = CONTINUE$	0.55	0.81	1.63	0.84	0.60	1.31
$H_0: START = EXIT$	4.28**	3.74*	3.89**	4.33**	3.71*	3.98**
H_0 : CONTINUE = EXIT	3.48*	6.54**	7.77***	3.17*	6.25**	7.60***
Sector and Country Fixed Effects	No	Yes	No	No	Yes	No
Sector-Country Fixed Effects	No	No	Yes	No	No	Yes
Observations	2,125	2,125	2,125	2,125	2,125	2,125
F-Statistic	72.08***	24.13***	5.05***	72.07***	25.43***	5.36***
R-squared	0.254	0.372	0.459	0.273	0.390	0.475
Robust standard errors in parenthes	es					
*** p<0.01, ** p<0.05, * p<0.1						

Finally, we turn to the regression approach to search for productivity differences between export starters, continuers and exiters. Results from OLS regressions are reported in Table 9 with quantile regression results reported in Table 10. Concentrating on the exporter variables we observe that across the different specifications there is an insignificant (negative) coefficient on the export exiter dummy. As such, export exiters are not found to perform differently to non-exporters. For export continuers we find coefficients that are consistently large, positive and significant. In the case of export starters we obtain consistently positive coefficients that are in half of the cases are significant. The coefficients tend to be significant until country-sector fixed effects are included. Given these results we would be tempted to conclude that export continuers perform better than export starters, a result that would point

towards learning-by-exporting effects. Our results further indicate however that there are no significant differences between the coefficients for export starters and export continuers. For the comparison between both continuers and exiters and between starters and exiters however, we do consistently observe significant differences in the coefficients. When considering the quantile regression results (Table 10) we again find large, positive and significant coefficients on the dummy variables for export starters and continuers (except at the highest quantiles), though the coefficients tend to be larger for export continuers. The coefficients for exiters are generally insignificant across quantiles, though at the lowest quantile (10th percentile) the coefficient is actually negative and significant. Despite these differences in coefficients, few are found to be significant. Only at the lowest percentile do we find significant differences in coefficients in the case of both starters and exiters, and continuers and exiters. The result that continuers perform better than exiters, for some firms at least, is consistent with results found by Girma et al. (2003) for manufacturing firms who argue that such results imply that domestic output does not compensate for the loss of foreign market share.

Table 10					
Quantile Re	gression Resu	Ilts for Export	Starters, Cont	inuers and Exi	ters
	(1) 10 th	(2) 30 th	(3) 50 th	(4) 70 th	(5) 90 th
ln <i>EMP</i>	0.325*	0.515***	0.427***	0.258**	0.195
	(0.182)	(0.0956)	(0.132)	(0.108)	(0.189)
ln <i>EMP</i> ²	-0.0528**	-0.0714***	-0.0624***	-0.0449***	-0.0442*
	(0.0237)	(0.0121)	(0.0166)	(0.0135)	(0.0236)
AGE	0.0155***	0.0139***	0.0121***	0.0114***	0.0108***
	(0.00486)	(0.00246)	(0.00319)	(0.00249)	(0.00403)
KL	0.395***	0.334***	0.340***	0.314***	0.268***
	(0.0383)	(0.0179)	(0.0239)	(0.0198)	(0.0364)
HK	0.00676***	0.00898***	0.0106***	0.0111***	0.0111***
	(0.00194)	(0.00103)	(0.00144)	(0.00117)	(0.00185)
START	0.795*	0.795***	0.786**	0.362	0.387
	(0.470)	(0.259)	(0.346)	(0.280)	(0.407)
CONTINUE	1.013***	0.828***	0.478**	0.625***	0.483
	(0.257)	(0.165)	(0.232)	(0.191)	(0.308)
EXIT	-2.828***	0.191	0.433	0.119	-0.304
	(0.565)	(0.348)	(0.490)	(0.384)	(0.550)
IMP	0.505***	0.614***	0.675***	0.525***	0.398***
	(0.129)	(0.0712)	(0.0971)	(0.0783)	(0.125)
TWO	-0.0986	0.118	0.560**	0.645***	0.806**
	(0.322)	(0.195)	(0.270)	(0.223)	(0.345)
FOREIGN	0.339***	0.462***	0.398***	0.502***	0.610***
	(0.124)	(0.0676)	(0.0929)	(0.0758)	(0.122)
$H_0: START = CONTINUE$	0.26	0.02	0.73	0.82	0.04
H_0 : $START = EXIT$	24.81***	1.96	0.35	0.27	1.06
H_0 : $CONTINUE = EXIT$	39.27***	2.78*	0.01	1.41	1.59
Observations	2,260	2,260	2,260	2,260	2,260
Psuedo R ²	0.180	0.185	0.170	0.155	0.123
Standard errors in parentheses	5				
*** p<0.01, ** p<0.05, * p<0.1					

5. Conclusions

This paper adds to the firm-level literature on international trade and performance by using a recent and broad survey of firms in a large number of SSA countries. It provides the first evidence on the relationship between a firm's international trade status and its performance for firms in the services sector in SSA and as such is one of the first for developing countries more generally. Using a variety of parametric and non-parametric statistical techniques we find that services firms that are engaged in international trade perform significantly better than those firms that sell and purchase in the domestic market only. The estimated premium from engaging in trade is usually found to be large, and in the case of two-way traders often exceeds 100 per cent. As found elsewhere the productivity premium is found to differ for exporters, importers and two-way traders, though the pattern found in the current paper differs from that found in other studies considering mainly manufacturing firms. In particular, rather than observing that exporters are the worst performing of all trading firms, we observe that both two-way traders and exporters perform better than importers. Indeed, for some performance criteria we find that exporters only also outperform two-way traders.

The data available – which is a single cross-section – are unable to say much on the issue of whether these differences arise due to self-selection or learning-by-trading in SSA. It is hoped that future African Investor Surveys could be combined with the current survey to provide a panel dataset with time-series variation that could help us address this question more fully as well as control for firm heterogeneity more adequately. The data do allow us to examine productivity differences between export starters, continuers and exiters however. Results suggest that exiters perform less well than starters and continuers. The coefficients on continuers tend to be larger than those for starters, which is a result consistent with learning-by-exporting. Despite the coefficients often being quite different in terms of their size, there are no significant differences between the two coefficients however. This weakens the evidence in favour of learning-by-exporting, though the lack of significant differences may reflect the small number of export starters in the dataset. Yet, we can conclude that there is evidence of self-selection into exporting, since export starters outperform non-exporters, and that there may be some limited evidence in favour of learning-byexporting. Self-selection suggests that policies that enhance firms' competitiveness should be pursued. In addition, to the extent that learning-by-exporting is present our results would suggest that policies encouraging international exposure of services firms – for exporters in particular – should be promoted.

References

Andersson, M., Lööf, H. and S. Johansson, 2008. Productivity and international trade: Firm level evidence from a small open economy. *Review of World Economics*, 144, 774-801.

Antras, P. and E. Helpman, 2004. Global sourcing. CEPR Discussion Paper no. 4170, Centre for Economic Policy Research.

Aw, B.Y., Chung, S. and M.J. Roberts, 2000. Productivity and turnover in the export market: Micro-level evidence from Taiwan (China) and the Republic of Korea. *World Bank Economic Review*, 14, 65-90.

Bernard, A.B. and J.B. Jensen, 1995. Exporters, jobs, and wages in U.S. manufacturing: 1976-1987. *Brookings Papers on Economic Activity: Microeconomics*, 67-119.

Bhattacharya, R., Patnai, I. and A. Shah, 2010. Exports versus FDI in services. IMF Working Paper no. 290, International Monetary Fund, Washington: DC.

Bigsten, A., Collier, P., Dercon, S., Fafchamps, M., Gauthier, B., Gunning, J.W., Oduro, A., Oostendorp, R., Pattillo, C., Soderbom, M., Teal, F. and A. Zeufack, 2004. Do African manufacturing firms learn from exporting. *Journal Development Studies*, 40, 115-141.

Buchinsky, M., 1998. Recent advances in quantile regression methods: A practical guideline for empirical research. *Journal of Human Resources*, 33, 88-126.

Canay, I., 2011. A simple approach to quantile regression for panel data. Econometrics Journal, 14, 368-386.

Castellani, D., Serti, F. and C. Tomasi, 2010. Firms in international trade: Importers' and exporters' heterogeneity in the Italian manufacturing industry. *World Economy*, 33, 424-457.

Coe, D. T. and E. Helpman, 1995. International R&D Spillovers. European Economic Review, 39, 859-887.

Coe, D. T., E. Helpman and A. W. Hoffmaister, 1997. North-South R&D spillovers. *Economic Journal*, 107, 134-149.

Delgado, M.A., Farinas, J.C. and S. Ruano, 2002. Firm productivity and export markets: A non-parametric approach. *Journal of International Economics*, 57, 397-422.

Foster-McGregor, N., Isaksson, A. and F. Kaulich, 2012. Importing, exporting and performance in sub-Saharan African manufacturing firms. Mimeo.

Francois, J. and B. Hoekman, 2010. Services trade and policy. Journal of Economic Literature, 48, 642-692.

Fryges, H. and J. Wagner, 2007. Exports and productivity growth – first evidence from a continuous treatment approach. Discussion Paper no. 07-032, Centre for European Economic Research.

Harris, R., 2002. Foreign ownership and productivity in the United Kingdom – Some issues when using the ARD establishment level data. *Scottish Journal of Political Economy*, 49, 318-335.

Harris, R. and Q. Li, 2007. Firm level empirical study of the contribution of exporting to UK productivity growth. Report to UK Trade and Investment.

Harris, R. and C. Robinson, 2003. Foreign ownership and productivity in the United Kingdom: Estimates for UK manufacturing using the ARD. *Review of Industrial Organisation*, 22, 207-223.

Koenker, R., 2004. Quantile regression for longitudinal data. *Mimeo*. University of Illinois.

Koenker, R. and K. Hallock, 2001. Quantile regression. Journal of Economic Perspectives, 15, 143-156.

Kraay, A., 2002. Exports and economic performance: Evidence from a panel of Chinese enterprises. In Mary-Francoise Renard (ed.), *China and its Regions. Economic Growth and Reform in Chinese Provinces*, Cheltenham, Edward Elgar, 278-299. (Originally published in Revue d'Economique du Developpement, 1-2, 1999).

Martins, P. and Y. Yang, 2009. The impact of exporting on firm productivity: A meta-analysis of the learning-by-exporting hypothesis. Review of World Economics, 145, 431-445.

Melitz, M.J., 2003. The impact of trade on intra-industry reallocations and aggregate industry productivity. *Econometrica*, 71, 1695-1725.

Massimiliano, C., Ellis, K. and D.W. te Velde, 2008. The contribution of services to development: The role of regulation and trade liberalisation. London, Overseas Development Institute.

Muuls, M. and M. Pisu, 2009. Imports and exports at the level of the firm: Evidence from Belgium. *World Economy*, 32, 692-734.

Powell, D., 2010. Unconditional quantile regression for panel data with exogenous or endogenous regressors. The selected works of David Powell, available at: http://works.bepress.com/david_powell/1

Powell, D. and J. Wagner, 2011. The exporter productivity premium along the productivity distribution: Evidence from unconditional quantile regression with firm fixed effects. RAND Working Paper no. 837, RAND Corporation.

Triplett, J. and B. Bosworth, 2004. *Productivity in the US Services Sector: New Sources of Economic Growth.* Brookings Institution Press, Washington: D.C.

UNIDO, 2012. Africa Investor Report 2011: Towards Evidence-Based Investment Promotion Strategies. Vienna, United Nations Industrial Development Organization.

Vogel, A. and J. Wagner, 2010. Higher productivity in importing German manufacturing firms: Self-selection, learning from importing, or both? *Review of World Economics*, 145, 641-665.

Wagner, J., 2007. Exports and productivity: A survey of the evidence from firm-level data. *World Economy*, 30, 60-82.

Wagner, J., 2012. International trade and firm performance: A survey of empirical studies since 2006. *Review of World Economics*, 148, 235-267.

WTO, 2008. Statistics Database, International Trade and Tariffs Data. Geneva, World Trade Organisation.

Yasar, M. and C.J. Morrison Paul, 2007. International linkages and productivity at the plant level: Foreign direct investment, exports, imports and licensing. *Journal of International Economics*, 72, 373-388.

Appendix

Table A1

Number of Firms in Sample by Country

Country	No. of Firms (% of Total)	Exporters	Importers	Two-way Trad- ers	Foreign-owned
Burkina Faso	43 (1.89)	4	8	0	6
Burundi	96 (4.22)	3	18	3	24
Cameroon	125 (5.49)	11	33	13	67
Cape Verde	232 (10.19)	4	83	6	63
Ethiopia	101 (4.44)	7	37	3	22
Ghana	117 (5.14)	5	44	7	59
Kenya	197 (8.65)	18	35	23	49
Lesotho	67 (2.94)	0	41	2	16
Madagascar	99 (4.35)	9	19	6	54
Malawi	42 (1.84)	0	18	2	18
Mali	113 (4.96)	6	28	7	43
Mozambique	111 (4.87)	2	87	2	38
Niger	37 (1.62)	1	11	0	10
Nigeria	130 (5.71)	0	27	3	19
Rwanda	67 (2.94)	3	20	5	36
Senegal	128 (5.62)	9	28	11	51
Tanzania	119 (5.23)	5	31	4	42
Uganda	348 (15.28)	19	100	24	173
Zambia	105 (4.61)	5	41	7	27
Total	2,277 (100)	111	709	128	817

Table A2

Number of Firms in Sample by Industry

Industry	No. of Firms (% of Total)	Exporters	Importers	Two-Way Traders	Foreign-Owned
Sale, maintenance and repair of motor vehicles and motorcycles; retail sale of automotive fuel	202 (8.87)	8	91	17	64
Wholesale trade and commission trade, except of motor vehicles and motorcycles	353 (15.50)	19	193	49	151
Retail trade, except of motor vehicles and motorcycles; repair of personal and household goods	333 (14.62)	8	195	23	105
Hotels and restaurants (281), Land transport; transport via pipelines	281 (12.34)	4	61	0	87
Land transport; transport via pipelines	140 (6.15)	13	30	10	39
Water transport	22 (0.97)	2	5	2	14
Air transport	19 (0.83)	1	2	3	11
Supporting and auxiliary transport activities; activities of travel agencies	85 (3.73)	7	5	2	25
Post and telecommunications	80 (3.51)	6	20	7	44
Financial intermediation, except insurance and pension funding	170 (7.47)	7	9	1	87
Insurance and pension funding, except compulsory social security	86 (3.78)	6	3	1	38
Activities auxiliary to financial intermediation	21 (0.92)	0	0	0	5
Real estate activities	66 (2.90)	0	10	0	21
Renting of machinery and equipment without operator and of personal and household goods	16 (0.70)	0	3	1	5
Computer and related activities	37 (1.62)	4	9	6	14
Research and development	1 (0.04)	0	0	0	0
Other business activities	269 (11.81)	23	45	5	88
Public administration and defence; compulsory social security	4 (0.18)	1	0	0	0
Education	23 (1.01)	0	1	0	8
Health and social work	11 (0.48)	0	3	0	2
Sewage and refuse disposal, sanitation and similar activities	32 (1.41)	2	12	0	2
Recreational, cultural and sporting activities	18 (0.79)	0	7	1	6
Other service activities	7 (0.31)	0	5	0	1
Activities of private households as employers of domestic staff	1 (0.04)	0	0	0	0
Total	2,277 (100)	111	709	128	817

Short list of the most recent wiiw publications (as of March 2013)

For current updates and summaries see also wiiw's website at www.wiiw.ac.at

Importing, Exporting and the Productivity of Services Firms in Sub-Saharan Africa

by Neil Foster-McGregor, Anders Isaksson and Florian Kaulich

wiiw Working Papers, No. 98, March 2013 26 pages including 12 Tables

hardcopy: EUR 8.00 (PDF: free download from wiiw's website)

Outward Foreign Direct Investment, Exporting and Firm-Level Performance in Sub-Saharan Africa

by Neil Foster-McGregor, Anders Isaksson and Florian Kaulich

wiiw Working Papers, No. 97, March 2013 23 pages including 9 Tables and 9 Figures

hardcopy: EUR 8.00 (PDF: free download from wiiw's website)

Importing, Exporting and Performance in sub-Saharan African Manufacturing Firms

by Neil Foster-McGregor, Anders Isaksson and Florian Kaulich

wiiw Working Papers, No. 96, March 2013

25 pages including 12 Tables

hardcopy: EUR 8.00 (PDF: free download from wiiw's website)

Double-dip Recession over, yet no Boom in Sight

by Vasily Astrov, Vladimir Gligorov, Doris Hanzl-Weiss, Mario Holzner, Michael Landesmann, Olga Pindyuk et al.

wiiw Current Analyses and Forecasts. Economic Prospects for Central, East and Southeast Europe, No. 11, March 2013

157 pages including 31 Tables and 35 Figures hardcopy: EUR 80.00 (PDF: EUR 65.00)

Does Trade Drive Global Growth?

by Leon Podkaminer

wiiw Research Reports, No. 386, March 2013 19 pages including 4 Tables and 6 Figures hardcopy: EUR 15.00 (PDF: EUR 8.00)

wiiw Monthly Report 3/13

edited by Leon Podkaminer

- · Political turmoil and early elections in Bulgaria
- A revival of industrial policy as a growth model for Russia?
- The determinants of income polarization at the household and country level across the EU
- Statistical Annex: Selected monthly data on the economic situation in Central, East and Southeast Europe

wiiw, March 2013

22 pages including 10 Tables

(exclusively for subscribers to the wiiw Service Package)

What is Happening to Growth in Europe?

by Rumen Dobrinsky

wiiw Research Reports, No. 385, February 2013 57 pages including 8 Tables and 22 Figures hardcopy: EUR 24.00 (PDF: EUR 15.00)

R&D and Non-R&D Innovators in the Financial Crisis: the Role of Binding Credit Constraints by Sandra M. Leitner and Robert Stehrer

wiiw Working Papers, No. 95, February 2013 31 pages including 12 Tables and 3 Figure hardcopy: EUR 8.00 (PDF: free download from wiiw's website)

wiiw Monthly Report 2/13

edited by Leon Podkaminer

- The new Multiannual Financial Framework of the European Union for 2014-2020: a comment
- · Labour costs, external competitiveness and economic growth in new EU member states
- Global values new reflections and new data on an old debate
- Statistical Annex: Selected monthly data on the economic situation in Central, East and Southeast Europe

wiiw, February 2013

32 pages including 15 Tables, 3 Figures and 4 Maps (exclusively for subscribers to the wiiw Service Package)

Whither Panama? Constructing a Consistent and Balanced World SUT System including International Trade and Transport Margins

by Gerhard Streicher and Robert Stehrer

wiiw Working Papers, No. 94, January 2013 21 pages including 5 Tables and 3 Figure hardcopy: EUR 8.00 (PDF: free download from wiiw's website)

wiiw Monthly Report 1/13

edited by Leon Podkaminer

- The European Commission grossly underestimates the effects of German domestic demand expansion
- The new North-South Divide in Europe can the European convergence model be resuscitated?
- Changes in the European convergence model
- Statistical Annex: Selected monthly data on the economic situation in Central, East and Southeast Europe

wiiw, January 2013

32 pages including 8 Tables and 11 Figures (exclusively for subscribers to the wiiw Service Package)

wiiw Monthly Report 12/12

edited by Leon Podkaminer

- Draghi's démarche and the nightmare of the Bundesbank
- · Trade does not drive global growth
- · A note on Kazakhstan's Oil Fund
- Statistical Annex: Selected monthly data on the economic situation in Central, East and Southeast Europe

wiiw, December 2012

28 pages including 13 Tables and 5 Figures

(exclusively for subscribers to the wiiw Service Package)

The Role of Multinational Companies in International Business Integration

by Gábor Hunya

wiiw Research Reports, No. 384, November 2012 43 pages including 8 Tables and 18 Figures hardcopy: EUR 8.00 (PDF: free download from wiiw's website)

wiiw Monthly Report 11/12

edited by Leon Podkaminer

- Eurozone crisis and fiscal austerity push half of the CESEE region into recession
- · Albania: EU, ho!
- Bosnia and Herzegovina: change without evolution
- Croatia: employment victimized by budget consolidation
- Kazakhstan: slower growth due to lower oil production
- · Macedonia: back to recession
- Montenegro: new old government
- · Russian Federation: growth flat, politics bad
- Serbia: new government, old problems
- Ukraine: incumbent political elite retains its grip
- Statistical Annex: Selected data on FDI in Central, East and Southeast Europe

wiiw, November 2012

40 pages including 18 Tables

(exclusively for subscribers to the wiiw Service Package)

Global Value Chains and the EU Industry

by Robert Stehrer, Marcin Borowiecki, Bernhard Dachs, Doris Hanzl-Weiss, Steffen Kinkel, Johannes Pöschl, Magdolna Sass, Thomas Christian Schmall and Andrea Szalavetz

wiiw Research Reports, No. 383, October 2012 152 pages including 42 Tables and 34 Figures

hardcopy: EUR 8.00 (PDF: free download from wiiw's website)

wiiw Service Package

The Vienna Institute offers to firms and institutions interested in unbiased and up-to-date information on Central, East and Southeast European markets a package of exclusive services and preferential access to its publications and research findings, on the basis of a subscription at an annual fee of EUR 2,000.

This subscription fee entitles to the following package of **Special Services**:

- A free invitation to the Vienna Institute's Spring Seminar, a whole-day event at the end of March, devoted to compelling topics in the economic transformation of the Central and East European region (for subscribers to the wiiw Service Package only).
- Copies of, or online access to, *The Vienna Institute Monthly Report*, a periodical consisting of timely articles summarising and interpreting the latest economic developments in Central and Eastern Europe and the former Soviet Union. The statistical annex to each *Monthly Report* contains, alternately, country-specific tables or graphs with monthly key economic indicators, economic forecasts, the latest data from the wiiw Industrial Database and excerpts from the wiiw FDI Database. This periodical is not for sale, it can only be obtained in the framework of the wiiw Service Package.
- Free copies of the Institute's Research Reports (including Reprints), Current Analyses and Forecasts, Country Profiles and Statistical Reports.
- A free copy of the wiiw Handbook of Statistics, published in October/November each year and containing macroeconomic and structural indicators of Albania, Bosnia and Herzegovina, Bulgaria, Croatia, the Czech Republic, Estonia, Hungary, Latvia, Lithuania, Macedonia, Montenegro, Poland, Romania, Russia, Serbia, the Slovak Republic, Slovenia and Ukraine
- Free online access to the wiiw Monthly Database, containing more than 2000 leading indicators monitoring the latest key economic developments in ten Central and East European countries.
- Consulting. The Vienna Institute is pleased to advise subscribers on questions concerning
 the East European economies or East-West economic relations if the required background
 research has already been undertaken by the Institute (10 hours p.a.).
- Free access to the Institute's specialised economics library and documentation facilities.

There will be a new, Premium Service Package available starting in January 2014 that includes, in addition to the standard package, online access to the wiiw Annual Database and wiiw FDI Database, as well as all country forecast tables in Excel format.

For detailed information about the wiiw Service Packages please visit wiiw's website at www.wiiw.ac.at

To The Vienna Institute for International Economic Studies Rahlgasse 3 A-1060 Vienna

- O Please forward more detailed information about the Vienna Institute's Service Package
- O Please forward a complete list of the Vienna Institute's publications to the following address

Please enter me for

- 1 yearly subscription of Research Reports (including Reprints) at a price of EUR 120.00 (hardcopy, Austria),
 EUR 135.00 (hardcopy, Europe), EUR 155.00 (hardcopy, overseas) and EUR 48.00 (PDF download with password) respectively
- 1 yearly subscription of Current Analyses and Forecasts at a price of EUR 150.00 (hardcopy, Austria),
 EUR 155.00 (hardcopy, Europe), EUR 170.00 (hardcopy, overseas) and EUR 120.00 (PDF download with password) respectively
- O 1 combination of both yearly subscriptions (Current Analyses and Forecasts and Research Reports) at a price of EUR 225.00 (hardcopy, Austria), EUR 250.00 (hardcopy, Europe), EUR 265.00 (hardcopy, overseas) and EUR 140.00 (PDF download with password) respectively

Pie	ase forward						
0	the following issue of Research Reports						
0	the following issue of Current Analyses and Forecasts						
0	the following issue of Working Papers						
0	the following issue of Research Papers in German language						
0	the following issue of wiiw Database on Foreign Direct Investment						
0	the following issue of wiiw Handbook of Statistics						
0	(other)						
 Nai	~~						
ivai	ne						
••••							
Add	dress						
Tel	ephone Fax E-mail						
Dat	re Signature						

Herausgeber, Verleger, Eigentümer und Hersteller:

Verein "Wiener Institut für Internationale Wirtschaftsvergleiche" (wiiw),

Wien 6, Rahlgasse 3 ZVR-Zahl: 329995655

Postanschrift: A-1060 Wien, Rahlgasse 3, Tel: [+431] 533 66 10, Telefax: [+431] 533 66 10 50

Internet Homepage: www.wiiw.ac.at

Nachdruck nur auszugsweise und mit genauer Quellenangabe gestattet.

P.b.b. Verlagspostamt 1060 Wien